Author's response to reviews

Title: Prognostic significance of Hypoxia-Inducible Factor 1 alpha (HIF-1alpha) expression in serous ovarian cancer: an immunohistochemical study

Authors:

Alexandros Daponte (dapontea@otenet.gr)
Maria Ioannou (mioan@med.uth.gr)
Ilias Mylonis (simos@med.uth.gr)
George Simos (simos@med.uth.gr)
Marcos Minas (markosminas@hotmail.com)
Ioannis E Messinis (messinis@med.uth.gr)
George Koukoulis (poroi@otenet.gr)

Version: 6 Date: 13 August 2008

Author's response to reviews:

Dear Editor,

The revised manuscript was edited as suggested in your e-mail. Table 2 corrected, References and figures are in the order they appear in the text. Please contact me if further changes are needed.

Sincerely yours

Alexandros Daponte MD FCOG